

# Human-centred design process when users have an evolving profile: a case study in Parkinson's disease domain

KATHIA M OLIVEIRA<sup>\*</sup>, Université Polytechnique Hauts-de-France, France CHRISTOPHE KOLSKI, Univ. Polytechnique Hauts-de-France, France SOPHIE LEPREUX, Université Polytechnique - Hauts de France, France JULIA GREENFIELD, Université Polytechnique Hauts-de-France, France VÉRONIQUE DELCROIX, UPHF, France YOHAN GUERRIER, Université Polytechnique des Hauts-de-France, France ELISE BATSELÉ, University of Mons, Belgium TAISA GUIDINI GONÇALVES, Université Polytechnique Hauts-de-France, France PHILIPPE PUDLO, Université Polytechnique Hauts-de-France, France ROMINA RINALDI, University of Mons, Belgium ANNE BLANCHARD-DAUPHIN, Lille University Hospital Center, France

**Abstract**. The human-computer interaction community has largely defended the use of an iterative process in which the users and stakeholders participate in all activities throughout the development of an interactive system. This work can become challenging when users have an evolving profiles in degenerative diseases (e.g. Parkinson's and Alzheimer's). The needs of the users at the start of the process may no longer be the same at the end when the system is validated. As consequence, it is important to integrate users in different stages of the disease in the whole process. The involvement of stakeholders (such as healthcare professionals, family members and caregivers) is also crucial for the better understanding of the user's needs and their evolving profile. This article presents how we deal with the evolving user profile in the development of a communication aid software for people with Parkinson's disease. To that end, we have customized the ISO 9241-210 human-centred process and the complementary studies carried out with 97 people with Parkinson's disease, 5 relatives/caregivers and 5 health professionals. In light of the insights gained from this experience, we have formulated several recommendations for human-centred design in the context of a user with an

\*corresponding author, kathia.oliveira@uphf.fr

© 2025 Copyright held by the owner/author(s). Publication rights licensed to ACM.

ACM 2573-0142/2025/6-ARTEICS019

https://doi.org/10.1145/3735500

Authors' Contact Information: Kathia M Oliveira, Université Polytechnique Hauts-de-France, LAMIH UMR CNRS 8201, Valenciennes, France, kathia.oliveira@uphf.fr; Christophe Kolski, Univ. Polytechnique Hauts-de-France, LAMIH UMR CNRS 8201, Valenciennes, France, christophe.kolski@uphf.fr; Sophie Lepreux, Université Polytechnique - Hauts de France, LAMIH UMR CNRS 8201, Valenciennes, France, sophie.lepreux@uphf.fr; Julia Greenfield, Université Polytechnique Hautsde-France, LAMIH UMR CNRS 8201, Valenciennes, France, julia.greenfield@hotmail.co.uk; Véronique Delcroix, UPHF, LAMIH UMR CNRS 8201, Valenciennes, France, veronique.delcroix@uphf.fr; Yohan Guerrier, Université Polytechnique des Hauts-de-France, LAMIH UMR CNRS 8201, Valenciennes, France, yohan.guerrier@uphf.fr; Elise Batselé, University of Mons, Mons, Belgium, elise.batsele@umons.ac.be; Taisa Guidini Gonçalves, Université Polytechnique Hauts-de-France, LAMIH UMR CNRS 8201, VALENCIENNES, France, taisaguidini@gmail.com; Philippe Pudlo, Université Polytechnique Hauts-de-France, LAMIH UMR CNRS 8201, Valenciennes, France, philippe.pudlo@uphf.fr; Romina Rinaldi, University of Mons, Mons, Belgium, romina.rinaldi@umons.ac.be; Anne Blanchard-Dauphin, Lille University Hospital Center, Physical Medicine and Rehabilitation Service, Lille, France, anne.blanchard@chu-lille.fr.

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than the author(*s*) must be honored. Abstracting with credit is permitted. To copy otherwise, or republish, to post on servers or to redistribute to lists, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

evolving profile. These recommendations outline a human-centred approach to guide the development of software applications for users with evolving profiles.

# $\label{eq:ccs} \texttt{CCS Concepts:} \bullet \textbf{Software and its engineering} \rightarrow \textbf{Software design engineering}; \bullet \textbf{Human-centered computing} \rightarrow \textbf{User studies}.$

Additional Key Words and Phrases: human-centred design, user-centred design, empirical studies, evolving user profile, people with Parkinson's disease

# **ACM Reference Format:**

Kathia M Oliveira, Christophe Kolski, Sophie Lepreux, Julia Greenfield, Véronique Delcroix, Yohan Guerrier, Elise Batselé, Taisa Guidini Gonçalves, Philippe Pudlo, Romina Rinaldi, and Anne Blanchard-Dauphin. 2025. Human-centred design process when users have an evolving profile: a case study in Parkinson's disease domain. *Proc. ACM Hum.-Comput. Interact.* 9, 4, Article EICS019 (June 2025), 34 pages. https://doi.org/10.1145/3735500

# **1 INTRODUCTION**

User-centred design has been used in information technology since the 1980s/1990s, arguing that placing the user at the centre of system design, considering usability issues, is fundamental to the success of a project [56, 66]. In 1999, ISO calls this process human-centred design<sup>1</sup>, to emphasize the importance of considering various stakeholders in the process and not just typical users [38]. Much work has been developed for different application domains considering different user profiles, such as children [46, 80, 83], the elderly [27, 39, 77] or people with disabilities [10, 12, 30, 31, 79]. We argue that applying this approach is necessary, yet not trivial while dealing with users with an evolving profile.

A profile is considered as evolving when certain characteristics of the user evolve progressively or abruptly over time in a degenerative manner (for example, progressive loss of movement/voice for people with Parkinson's, or memory for people with Alzheimer's). In this case, it is important to continue identifying the user needs throughout different stages of the disease in order to provide the right software for the desired purpose. We faced this situation when working on the European project named ParkinsonCom<sup>2</sup> [21], which aimed to develop a communication aid software for People with Parkinson's disease (PwP) in order to improve their social participation and inclusion. At the beginning of the project several questions emerged, such as: What are the communication needs of a person with Parkinson's disease? What are their difficulties? What is the adequate kind of interaction (e.g.: gestures, vocal, touchscreen)? What is the adequate human-computer interaction to provide? These questions are very common in any kind of system and particularly important to be well-addressed when the users are people with disabilities. Moreover, in this context, we had to deal with the degenerative character of the disease, and consequently the degeneration of the ability to communicate. This meant that we had to understand the different communication needs at each moment of the disease and that this would probably imply different technological support for communication. In summary, our primary research question is as follows: How can we design an interactive system for people with an evolving profile?.

In order to respond to the this research question, we advocate two key points. First, all stakeholders (users, medical professionals, caregivers, etc.) should be included in the core of the design process. This means that a human-centred design should be considered. Second, it is imperative to address each level of the evolving profile in a manner that aligns with its specific needs. Based on these assumptions, the goal of this work is to customize the ISO 9241-210 [38] standard for the development of the aforementioned interactive system for Parkinson's disease domain. All the activities in this process are described taking into account the particularities of dealing with

<sup>&</sup>lt;sup>1</sup>human-centred design processes for interactive systems in the withdrawn ISO 13407 standard revised by [38].
<sup>2</sup>https://parkinsoncom.eu

Proc. ACM Hum.-Comput. Interact., Vol. 9, No. 4, Article EICS019. Publication date: June 2025.

Parkinson's evolving profile integrating several human-computer studies. Based on the lessons learned from this experience, we have taken a step forward, proposing a generic process for dealing with the development of interactive systems when the user has an evolving profile.

The contribution of this article is twofold:

- i. the definition and application of a human-centred design process for the development of a communication aid software for people with Parkinson's disease; and,
- ii. the proposal of a human-centred design process to be used when users have evolving profile. This process specifies all the activities to be carried out for the development of interactive systems taking into account the evolving user profile and the participation of the different stakeholders involved in the process.

This paper is further composed of four sections. Section 2 describes the study background, briefly introducing the user profile of PwP and presenting the related work on human-centred design studies. Section 3 presents the customization of the human-centred design process adapted from ISO 9241-210 [38] for the development of a communication tool for people with Parkinson's disease. Section 4 summarises our lessons learned in carrying out this process over almost three years and proposes a first version of a generic human-design process to be applied when users have an evolving profile. Section 5 presents our conclusions and future work prospects.

## 2 BACKGROUND

# 2.1 Evolving User Profile in Human-Computer Interaction

In the design of an interactive system, two principles complement each other. On the one hand, user-centered design approaches which are necessary to properly take into account the needs and capabilities of users in order to offer usable interactive systems and a good user experience. This research work resulted, among others, in ISO 9241-210 [38] standard. On the other hand, adaptive system design approaches, introduced in the 1990s, focused on transfers of interaction between platforms [64] to the point of integrating a whole set of adaptations. Calvary et al. [17] propose the adaptation of the human–computer interfaces to their context of use, seen as a triplet: <user, platform, environment> ; the user is the part which is the focus in that article. In order to integrate the user, the notion of profile was therefore put in place [24, 47, 53], and regularly enriched to cover increasingly broad profiles and particular characteristics such as user personality [3]. Other approaches use user models [2, 26] in order to propose more intelligent and more complete adaptations.

However, human beings evolve during the use of the system. The evolution may be in terms of competence in the use of the system, as illustrated in a groupware context [14]. The evolution can also concern the needs in terms of progression in professional skills. In the fields of education [59, 60] or games [13, 74], we will find for example level systems to evaluate this and propose coherent activities with the progress of the learner, resp. of the player. The user profile can also change over the course of their life when a deficiency or an illness occurs (e.g. [32, 71, 72]). The developments can be in the direction of a regression which forces the user interface to take into account the associated deficiencies. If the impairment is permanent, the principles of inclusion and accessibility are adapted, but if the impairment is fluctuating, this is no longer sufficient. It is in this context that this research, aimed at integrating this profile evolution into the design of interactive systems, is located. However, we have not found any work in the literature targeting this consideration of the evolving profile in the development process.

# 2.2 The Evolving Profile of People with Parkinson's Disease

Parkinson's disease (PD) is a progressive, multi-system neurodegenerative disease that mainly affects older people. It has been identified as the second most common neurodegenerative disease worldwide, after Alzheimer's disease [68]. This disease is characterised by the destruction of a specific population of neurons, namely the dopamine neurons, and a subsequent disruption of dopaminergic neurotransmission in the basal ganglia [62]. These neurons are involved in the control of body movements, the loss of which is the most characteristic motor symptom of this disease.

Among the motor symptoms, bradykinesia is noted in all cases, i.e., a slowness in the initiation of voluntary movements with a progressive reduction in the speed and amplitude of repetitive actions (hypokinesia). Bradykinesia is always accompanied by at least one additional motor symptom, such as muscular rigidity or rest tremor [68]. Motor symptoms have long been the focus of disease management although non-motor symptoms are often reported to be more bothersome for patients [61]. Non-motor symptoms are classified as: dysautonomic signs (such as, digestive disorders, swallowing problems, arterial hypotension, respiratory disorders), sensory disorders (pain), speech disorders and psychological and cognitive disorders (sleep disorders, memory disorders, fatigue, depression, anxiety, and apathy) [61].

As the disease progresses, there may be alternating periods in which the symptoms are largely under control thanks to treatment, and periods in which the treatment no longer has a sufficient effect. This is commonly called as an alternation between *ON* period, when the treatment is effective, and *OFF* period, when the treatment has no effect [76].

Several scales are used to diagnose Parkinson's disease and its stage. These include the UPDRS (Unified Parkinson Disease Rating Scale) [58], the Hoehn and Yahr scale [32], and the Parkinson's Well-Being Map<sup>TM</sup> [71]. The UPDRS is a widely applied rating instrument for PD composed of several items contributing to three subscales: (a) mentation, behavior, and mood; (b) activities of daily living; and (c) motor examination. The Hoehn and Yahr scale has also been used for the staging of the functional disability associated with Parkinson's disease. It helps to describe the progression of the disease through various stages, thus allowing to measure the severity of the case. Assessment with both scales requires a medical examination and the history given by the patients and their caregivers. The Parkinson's Well-Being Map<sup>TM</sup> has been designed by physicians and patients to make it easy for the patients themselves to describe their difficulties and highlight the most burdening symptoms in order to guide their care. The items, organised in different categories (Figure 1), are sentences that should be answered using a 5-point Likert scale (from 0-never to 4-always).

# 2.3 Human-centred Design in case of Users with Parkinson's Disease

Human-centred design, or user-centred design, is defined by ISO 9241-210 [38] as "an approach to interactive system development that aims to make systems usable and useful by focusing on users, their needs and requirements, and by applying human factors/ ergonomics, and usability knowledge and techniques". Although the terms user-centred and human-centred are often used synonymously [38], we prefer to use the term human-centred in this work because the participation of stakeholders directly related to the user with an evolving profile in the development of the application, is often necessary, as also defended by [30] for the design of interactive systems for people with disability, particularly in case of communication disorders.

The basis of this approach is that stakeholders and users are involved in the design, development and evaluation of the interactive system in a iterative process until the desired result, in a degree of usability, is achieved. Involvement during development enables user and task requirements to be identified and incorporated into the system specifications. This involvement also enables feedback

#### EICS019:5



Fig. 1. Categories of Well-Being Map<sup>™</sup> with the result of a self-assessment (red line)

to be obtained by testing the proposed design solutions. Indeed, users' participation is a source of knowledge about the context of use and the way in which they are likely to use the product. The context of use encompasses information about users, goals and tasks, resources, and environment (including the technical, physical, social, cultural and organizational environments) [37].

Human-centred design has been applied in different work on Parkinson's disease for different purposes. We can quote, for instance, for the detection of PD (e.g. [25, 44, 69, 82]); for self-monitoring, self-treatment or self-care of PwP (e.g. [20, 43, 72, 73]); for daily treatment monitoring (e.g. [40]), and for understanding communication disorders (e.g. [6, 50]), sleep and anxiety disorders (e.g. [22]), sleepiness, depressive mood, hallucination and psychosis (e.g. [23]). All these studies were focused on the disease in general considering diagnosis and monitoring and not taking into account the different stages or evolution of the disease or specifically of one of the aspects of the disease.

In terms of techniques, interviews [15, 57, 73, 82], workshops [49, 82], questionnaires [6, 15, 20, 43, 49, 50, 52, 73] and focus groups [20, 41] are usually applied in the user-centred process.

In previous work [29], we have carried out a rapid review [70] on human-centred design approaches for the development of communication software systems. In our research in Scopus (01/09/2020) with the search string (("user-centred" OR "human-centred" OR "user-centered" OR "human-centered" ) AND ("communication" OR "conversation" OR "ICT" AND "Parkinson"), no results were found. So we decided to use "OR Parkinson", to find studies on human-centred design for communication in any domain or the application of human-centred design for the Parkinson's domain. We excluded studies on robots. By doing the searching in Scopus followed with snowballing procedures, we found 15 works (e.g. [5, 20, 35, 48, 50, 51, 57, 75]) on the Parkinson's domain and 52 works on other domains such as: autistic spectrum disorder (e.g. [1, 18, 36, 80, 83]), cerebral palsy (e.g. [34]), elderly (e.g. [9, 27, 42, 77]), and intellectual disability (e.g. [19, 79]). None of the 15 papers on Parkinson's domain mentioned the use of the human-centred design standard (ISO9241-210 [38]) and only 5 of the 52 on the other domains cited its use or the previous version. Of these 5 articles, one [63] cites ISO 9241-210 only to clarify the definitions of usability, two others [1, 54] use the standard process detailing its activities in various phases, another [33] defines a three-phase process based on the standard's activities, and the last one [78] specifies the process for integrating stakeholders into the general process, but does not change the way the target user is taken into

account in the process. Furthermore, no work in this review took into account the evolving profile issues in their propositions. The studies always focus on a specific functionality considering the disability in general and not considering the evolution of the profile. Although this review was carried out in 09/2020 (when we started this research) and we have continued to work in this field and others related to people with special needs, as far as we know, this result is still valid, i.e., there is no work that takes into account the issues of the evolving profile in the design of interactive systems.

# 3 DESIGN AND IMPLEMENTATION OF AN INTERACTIVE SYSTEM FOR PEOPLE WITH PARKINSON'S DISEASE

As previously mentioned, we were confronted with the difficulty of dealing with a evolving user profile in the context of a project for Parkinson's disease domain. This Interreg France-Wallonie-Vlaanderen project was carried out between April 2020 and December 2022 [21]. The goal of this project was to develop a software aid to support PwP to communicate with their entourage. This project was conducted by two universities and a medium-sized enterprise. Moreover, two hospitals (Lille University Hospital Center, France and UZ Brussels Hospital, Belgium) and three Parkinson's Associations (FranceParkinson Nord in France; Vlaamse Parkinson Liga and APK Parkinson Association a.b.l. in Belgium) participated as collaborators.

To address the development of this software systems, we decided to work in the adaptation of the human-centred process of ISO 9241-210 standard [38] considering not only the users, but all other stakeholders involved in the problem. This adaptation was deemed necessary, as the standard process involves the identification of a set of user needs and subsequent iterative development with those needs as a focal point. In our particular case, we were confronted with a degenerative user profile, and it became evident that the communication requirements could vary significantly between stages of the Parkinson's disease. In essence, it was recognized that each stage of the disease's progression entailed the emergence of a novel user profile, characterised by distinct requirements. In essence, the process delineated in the ISO 9241-210 standard is being executed in its entirety for each identified level. After the identification of the various levels, the process is executed in multiple instances concurrently. The outcomes of these instances are then integrated into a unified system. Figure 2 presents this idea by the overlap of the three main activities of the process. We have decided that in the first activity we should carry out studies to identify the different levels of communication for Parkinson's disease we should have to address.

Moreover, we decide to integrate the main principle of participatory research [11] and participatory design [67] which advocates the active participation of the stakeholders in the process. The idea was to use the same activities as the standard, enriched with various human-centred studies, ranging from a *broad* to a *precise* vision for development and evaluation. For a broad vision, comprehensive studies should be carried out seeking to collect results from a significant number of users, thereby facilitating a general understanding of the problem. This can be achieved through the use of internet surveys, which provide a more extensive data set. This can be followed by in-depth interviews, which will facilitate the generation of more focused insights into a more precise understanding of the problem. Each activity was reviewed considering the perspective of developing an interactive system for users with an evolving profile, as described in the sub-sections presented bellow.

The activities presented in green in Figure 2 represents the execution of the evolving profile level that we will be present in detail in the next sub-sections, with particular attention paid to the integration of the evolving profile issues.

#### TIT Survey Plan the user-centred process Literature review Interview Context of us Understand and specify context of use iterate, if nteractive system tudy described in Section 3.2 necessary Usability that meets the principles requirements Specify the use Evaluate against the requirements requirements integra on and ré-evaluation Study described Study described in Section 3.5 in Section 3.3 Evaluation Protocol Functionalites Design and develop solutions to meet the requirements Legend evolving profile integration Study described and test Activit level 1 in Section 3.4 name evolving profile evolving profile level 2 level 3

Fig. 2. Human-centred design of an interactive system for PwP

# 3.1 Plan the Human-centred Process

As proposed in the ISO 9241-210 standard, we started the project planning the human-centred process for our project. In this activity we should describe the activities to be performed, the material and human resources, cost and time constraints.

Regarding human resources, besides the technical team, we identified the following stakeholders to participate in the process: a physician expert in Parkinson's disease, two neurologists, two speech therapists, and several caregivers from the Parkinson's associations. Moreover, the Parkinson's associations agreed to invite the PwP from the associations to participate in the project. In terms of material, at the beginning of the project we did not know what kind of technology could be adapted to PwP.

In order to have participatory research and design, we planned several studies from a broader perspective to a more precise one. We started (Section 3.2) with an initial interview with a physician and speech therapist and a general survey (Section 3.2.1) on PwP's difficulties and communication needs. Then we carried out interviews (Section 3.2.2) to better identify their needs and the different levels of the communication difficulties. Other interviews were also carried out for the requirements specification (Section 3.3) and to define the user interface design elements (Section 3.4) for the software application. Finally, after developing the application we planned evaluation sessions with PwP (Section 3.5). Threats to validity on these studies were duly analysed (Section 3.6).

We came up against two important restrictions in our project: the end of the project set to December 2022 (no extension would be accepted) and the COVID-19 pandemic which lasted from the start of the project in April 2020 until mid-2021 when the vaccines began to be used. This last restriction limited the interviews to video conference meetings and made it impossible to carry out participatory design for the conception of the user interfaces or the use of focus groups.

#### EICS019:7

# 3.2 Understand and Specify Context of Use

This activity was started by a literature review on PD, regarding symptoms, communication difficulties and treatments (the result of this literature review is presented in the background section, see Section 2.2).

With this study in mind, a first interview by video-conference with the physician expert in PD and a speech therapist took place. The interview was guided following five general questions:

- i. Given that Parkinson's disease is progressive, would it be possible to define which patient profile(s) we should focus on for the development of the communication aid software system?
- ii. What do you think would be interesting for PwP in terms of communication?
- iii. What kind of interaction with computers would be possible for PwP? Do you think we can combine the use of voice with touch (touching a screen, for example) and gesture (a specific movement) for communication?
- iv. What type of technological device do you think we could use for the software for this type of patient (e.g. cell phone, large-screen tablet, devices that can be worn by the patient as an ear kit, glasses with data display/augmented reality, etc.)?
- v. Do you know whether some patients are used to working with communication aids (if so, which ones)?

The interview took 45 minutes. Concerning the first question (i), they considered we should focus on people with oral communication problems, intelligibility problems and communication difficulties, if possible without cognitive impairment (no cognitive impairment, no cerebellar syndrome). This should be based on the difficulties encountered in daily life, where the disorders cause difficulty. The physician mentioned that the evolution of Parkinson's disease is mainly in terms of cognitive disorders. Also that assessments of articulation, speech capability and intensity are usually performed by speech therapists using specific tools and it is very difficult to do this assessment using software tools.

With respect to question (ii) they mentioned that PwP are moderately motivated. Thus, it is necessary to have something practical, easy to use and appropriate. According to the experts, if it is too elaborate, too difficult, PwP will not use it. "It has to be simple, intuitive and adaptive", mentioned the physician expert in PD.

On the kind of users' interaction (iii) they emphasized that PwP can point with their finger. But that oral communication tools are too long and difficult to use and that the use of gestures is not possible because PwP often have involuntary movements. They emphasized that any type of interaction that implies an immersive environment is not appropriate for PwP because it is too disturbing.

Regarding the potential technological devices (iv), they recall that most of the patients are elderly, over 60. Therefore, all advanced technologies (glasses, ear kit, virtual environment, etc.) are not possible for them. They suggest the use of tablet, half the size of an A4 page, with a virtual keyboard on one part of the screen and pictograms on the other (we represent that in Figure 2 including a picture of a tablet). They suggested also the use of different coloured vowels to make them easier to identify, and other elements of this kind to make it easier to use.

Finally, concerning the use of communication tools (v) they said that, in general, a small number of PwP have already worked with pictograms and alphabet books, but this is rare.

After this initial interview, a questionnaire for a survey was elaborated and reviewed by this physician. The details of this survey are presented in Section 3.2.1. To better understand the needs and identify the different levels of the evolving profile for PwP regarding communication, several interviews were carried out, as presented in Section 3.2.2.

*3.2.1 A Survey for Knowing PwP's User Profile and Needs.* This survey was designed to better understand the PwP's particularities, by identifying the different characteristics of PwP, their different needs, and their experience with technological support (computers, mobile phones, tablets, mouse, etc.). Our aim was to get as broad a view as possible, encompassing a large number of potential future users (PwP). For this reason, an online survey seemed appropriate as it provided a time-efficient way to collect data from a larger study sample and was feasible at the given time period with relevant restrictions (COVID-19 pandemic).

The research protocol for the investigation of PwP's user profile and needs covering a survey and interviews (presented in the next section) was approved by the ethical committee of the Faculty of Psychology and Educational Sciences <sup>3</sup>, University of Mons, Belgium.

*Goal.* To obtain an overview of the difficulties experienced in PD and the associated communication problems from the point of view of PwP and their relatives/caregivers. More specifically, our interest was to identify (i) the impact of PD on the daily lives of PwP, (ii) their communication difficulties, (iii) the strategies they use to communicate with their surroundings, and (iv) their needs in terms of social communication in everyday life. Moreover, in order to better define the type of technological device to use, we also looked to identify (v) known and used technological devices.

**Participants**. PwP and relatives representing a PwP. Our inclusion criteria therefore relate to the fact of being diagnosed as having developed PD or of being the close carer of a PwP. Specifically, we considered the response to questionnaires from PwP. The recruitment method was based on voluntary participation following the dissemination of information via the Internet and social networks.

**Methods.** A questionnaire was drawn up with identification information (age, gender, marital status, confirmation of diagnosis and approximate date of the diagnosis) and a set of closed questions to answer the five specific objectives defined: (*i*) *impact of PD symptoms on their life*, (*ii*) *communication difficulties*, (*iii*) *strategies used to aid communication*, (*iv*) *communication needs*, and (*v*) *the experience with technology devices*. Each of the closed questions also contained an open option (other) for free completion. The questionnaire was coded using LimeSurvey, taking precaution to guarantee the anonymity of the interviewees. The questionnaire was elaborated considering scientific literature and was reviewed by the physician specialised in PD previously interviewed.

**Results**. We received 56 complete responses to the questionnaires (35.71% from men (n=20) and 64.29% from women (n=36)). On average, respondents were 58 years old (standard deviation 11.9). Four early cases (less than 40 years old) were identified. We highlight the following results:

• With regard to the (*i*) *impact of PD symptoms* on their lives, the respondents were asked to answer on a 4-point Likert scale (no, minor, moderate or major -impact) the severity of 25 common symptoms. To analyse the answers we considered symptoms in two categories: physical (lower limb tremor, upper limb tremor, involuntary movements/dyskinesia, pain/cramps, pain in the joints of the upper limbs, pain in the joints of the lower limbs, slowness of movement, lack of precision in movement, difficulty initiating movement, difficulty walking, balance problems and/or falls, fatigue and/or sleep problems, and bowel and/or bladder problems) and psycho-social (sadness/negative thoughts/anxiety, hallucinations, feeling of loneliness, difficulty managing emotions, difficulty carrying out leisure activities, difficulty swallowing, difficulties with writing, difficulty communicating orally, and variations between *ON/OFF* periods). The three main physical symptoms with proven impacts (moderate or

<sup>&</sup>lt;sup>3</sup>No specific number was specified in the approval document dated 18/12/2020.

major impacts) are: fatigue and/or sleep problems (48.21%), pain/cramps (41.07%), slowness of movement (41.07%). The three main psycho-social symptoms with proven impacts (moderate or major impacts) are: difficulty concentrating (44.64%), difficulty writing (35.71%), difficulty communicating orally (33.93%).

- Concerning *(ii) communication difficulties*, they had to answer yes/no/"I don't know" to a list of 10 items (difficulty perceiving their own voice, communication-related fatigue, difficulty finding words, difficulty continuing a discussion, difficulty initiating a discussion, difficulty articulating, speech rate too slow or too fast, monotone voice, hoarse/harsh voice, and voice of low intensity). The three main difficulties most frequently encountered by respondents to the questionnaires were: voice of low intensity (71.43%), communication-related fatigue (60.71%), and difficulty finding words (58.93%).
- For the *(iii) strategies used to aid communication*, they should also answer yes/no/"I don't know" on a list of 11 items (avoiding difficult situations, waiting for a better moment to speak, shortening the information, speaking louder, preparing the talk in advance, informing others about the difficulties, speaking slower, asking for help, use of a communication software, use of pictograms/images, and writing instead of speaking). The strategies most frequently used by respondents to the questionnaires were (Figure 3): avoiding difficult situations (e.g. noisy, stressful) (80.36%), waiting for a better moment to speak (66.07%), and shortening the information, getting straight to the point (60.71%).
- Regarding the *(iv) communication needs*, the respondents were asked to rank situations (discussing a subject, making requests, expressing my feelings, exchanging ideas, giving an opinion, and telling a story) in order of importance for them. We scored the first classified event with six points and the last one with just one point. In this way, we obtained an average for the entire sample. The three main needs that emerged were: discussing a topic, making a request and expressing their feelings.
- Regarding (v) the experience with technology devices, the majority of respondents use computers (89.29%), smartphones (83.93%), tablets and touch screens (66.07%), for instant messaging (85.71%) and video conferencing systems (Messenger, Whatsapp, Skype, Viber, Duo, etc.) (85.71%). It is also noted that new interaction technologies such as virtual reality headset (7.14%), eyetracking (0%) and joysticks (3.57%) are indicated as mostly unused. This information is confirmed by medical experts in PD, who indicate that these technologies are not suitable for PwP.

Finally, in order to assess the impact of PD symptoms on respondents to the questionnaires with regard to the stage of the disease, an average was calculated based on the question related to the impact of PD symptoms on their daily lives, using the score attributed by each participant for physical and psychosocial symptoms, and communication difficulties. A correlation analysis showed no relationship between the stage of the disease (years since diagnosis) and the impact of physical symptoms (r= p.14, p= .29), psychosocial symptoms (r= .07, p=.60), and communication difficulties (r= .20, p= .13) meaning there is not a "typical" profile of linear evolution of symptoms with the stage of the disease being more advanced. Figure 4 visually confirms that the symptom impact varies between participants and that PwP having a more advanced disease do not always exhibit the most problematic symptoms/difficulties. The same correlation analysis was carried out with age and was found not significant for physical symptoms (r= .05, p= .68), psychosocial symptoms (r= .04, p= .74) and communication difficulties (r= -.03, p= .84) (see Figure 5).

These results confirmed that the communication is truly a difficulty for every PwP. To manage this difficulty, PwP tend to avoid situations requiring communication, but this is not always possible considering that communication is necessary for making simple requests or expressing feelings.

#### EICS019:11



Fig. 3. Communication strategies used by PwP



Fig. 4. Correlation analysis between the stage of the disease (years since diagnosis) and the impact of symptoms

Other compensatory strategies are also often used by the participants such as waiting for a better moment to speak or shortening the information to be communicated. In a smaller proportion, some PwP also indicated that they use communication software or pictograms/images or writing to help them to communicate (see Figure 3).

A correlation analysis highlighted that age is not linked with the use of technology (r=.10, p=.47) meaning that participants use technology as much at a young age as at an older one. The same was observed for the stage of the disease that showed no relationship with the use of technology (r=-.15, p=.26).

Thus, this study did not allow us to truly identify the different levels of the evolving profile to accurately target different communication support needs, as each PwP is affected differently. Indeed, the symptoms do not seem to be specifically influenced by the disease stage, or by age. Moreover, the use of technology is not directly linked with age reinforcing the idea that support



Fig. 5. Correlation analysis between age and the impact of symptoms

technology which targets communication could be a path to explore. We therefore decided to carry out interviews to delve deeper into these issues.

# 3.2.2 Interviews for Defining Evolving Profile Levels and Context of Use.

*Goal.* The interviews were planned with two main objectives: to better understand the physical and psycho-social impacts of the disease and their impact on communication (quantitatively identified in the survey presented in the previous section), and to define different levels for the evolving profile taking into account their communication difficulties.

**Participants**. PwP or relatives representing a PwP. The recruitment method was based on voluntary participation from members of the Parkinson's associations that collaborated with the project and by invitation disseminated on the Internet and social networks. The relatives were interviewed in case where the PwP could not express themselves orally. They were asked to answer considering the PwP persona as much as possible.

*Methods.* An interview guide was established starting with identification information (age, gender, marital status, confirmation of diagnosis and approximate date of diagnosis). Next, a set of questions covering the following topics was carried out: the interviewee's life journey (feelings about the disease and the diagnosis, daily activities and changes in activities/habits); the impact of PD on the interviewee's life from a motor and non-motor point of view; the difficulties and strategies used to communicate with the people around them; the interviewee's relationship with technology, and in particular their expectations regarding the creation of a technological tool to support communication. For anonymisation, each interviewee received a colour code.

**Results**. We carried out 14 interviews, corresponding to 10 men (71%) and 4 women (29%) with PD. Of these 14, 10 were carried out with the PwP and 4 with relatives (wife/husband), who act usually as caregivers, answering based on the day-to-day running of the PwP (three men and one woman). We note that the dates of diagnosis vary less than for respondents to the survey. For many, the diagnosis was made several years prior the interview date (average diagnosis received: 11 years and 10 months). The interviews took an average of one hour and were conducted by video-conference due to the COVID-19 pandemic. All interviews were transcribed and a qualitative analysis was performed related to the psycho-social perspective. This analysis can be found in [8].

In this paper, we focus on the needs of communication to define the different levels of the evolving profile. To that end we summarized the results of the interviews in the main elements collected in the interview, as shown in Table 1 for the interview of two PwP. Then, we summarized all communication needs (potential use for the system) that PwP would like to address in Table 2. We note that "make a request" or "communicate by text" are the most commonly cited communication needs.

The interviews confirmed what was observed in the survey, i.e., that PD symptoms are not directly related to the age or the number of years of diagnosis. However, by carrying out the interviews, we identified clearly that some participants communicate orally without problems (participants named Bordeaux, Red, Khaki and Cyan), some have some difficulty but can be understood (White, Pink, Black, Green, Beige), and some others with major difficulties or who are no longer able to speak, with the interview being conducted with their relatives (Brun, Gray, Violet and Ruby). For this last group, the need for assistance in "making a simple request" was expressed, contrasting to those who can still communicate and asked for a support tool to "find words in real time" while they are speaking.

Furthermore, all parties expressed that the *OFF* periods rendered them incapable of communicating effectively independent of the stage of the disease. They are aware that they simply need to wait for the *OFF* period to end.

A thorough analysis of the interviews revealed two distinct extremes in terms of the level of communication difficulty exhibited. On one end of the spectrum, participants demonstrated a capacity to articulate themselves with relative ease, exhibiting a near-normal level of verbal expression. On the other end, participants exhibited a complete absence of verbal communication (The interviews were conducted primarily with the relatives who were accompanying the PwP). Based on that, we have decided to use an ordinal scale to define the evolving profile level. This decision has been taken in light of the two extremes and a middle one. Consequently, three levels were delineated, as presented in Figure 6. Moreover, the scales for diagnosis of PD (see Section 2.2) usually have some items directly related to speaking ability. Even better, the Well-Being Map<sup>™</sup> scale (presented earlier, see Section 2.2), that is used for self-assessment of PwP, presents an item directly related to this ("I have trouble talking") and other items that can be associated to the problem of communication (e.g. "I have difficulty remembering names, numbers, events").



Fig. 6. Three levels of evolving profile for PwP related to communication (studies of level 3 are presented in this article)

Interview el- ements	Mr. Pink, 68 years old, PD since 2005	Mrs. Khaki, 61 years old, PD since 2016
Aspects of the disease	Sometimes his voice is dull, his dic- tion cottony and he has difficulty finding his words. <i>ON/OFF</i> periods.	Good communication, but problems with reversing or forgetting words, slow thinking. No variations between <i>ON/OFF</i> periods.
Communic. needs	Help and encourage people to express themselves in real time.	Being able to talk to her husband and friends. Conversing, expressing herself in meetings.
Daily activi- ties	Intellectual activity (writing a book), participation in several associations, corresponding with friends.	Frequent (sport, cultural activities, everyday activities, entertaining the family).
Moral	Good	Very good
Openness to technology	Wide open.	Moderate (good use of computer, smart- phone. Owns a tablet, but does not use it).
Ideas for technology support	Tablet or smartphone. Search for a word (in real time if possible). News and contextualised information.	A tool to help communicate with others PwP. Encourage and correct writing.
People sup- port	No	Mainly medical (neurologist every 6 months).
Strategies for commu- nic.	When he gets into a discussion too quickly, he waits for a better mo- ment.	Writing in preparation for meetings. Cutting off communication and wait for a favourable moment.

Table 1. Example of summary of interview answers from two People with PD

The interviews also revealed that future users would like a system mainly to be used in their own houses, to support daily life. They also mentioned that it could be interesting as support during medical visits. Therefore, two use environments should be taken into account as an element of the context of use: PwP's house and a doctor's office.

# 3.3 Specify the User Requirements

This activity was started by interviewing the physician expert in Parkinson's disease, the two neurologists, and a representative of the Parkinsons' associations. The goal was to present the results of the previous activity, the levels of PwP evolving profile defined for communication, and discuss potential solutions to the problem.

With the three levels identified and the list of needs, the interviewees scored which needs would be most relevant for each level. In addition, they acknowledged that user would not be able to use communication features during the *OFF* period independent of the evolving user profile level. However, some service that did not require a lot of interaction could be offered to help them wait

PwP colour code	Gender- Age	Make a request	Communi- cate by text (w/wo vocal synthesiser)	Find a word in real time	Network- ing (with PwP, doc- tors)	Increase voice volume	Know latest PD's news
Gray	W - 69	Х	Х				
Khaki	W - 61		Х		Х		
Blue	M - 60		Х	Х	Х		
White	W - 67		Х				Х
Black	M - 48			Х	Х		
Bor- deaux	M - 63						Х
Violet	M - 75	Х	Х			Х	
Pink	M - 68		Х	Х	Х		Х
Cyan	M - 71		Х				
Red	M - 64				Х		Х
Beige	M - 49		Х	Х	Х		
Ruby	M - 73		Х		Х		
Brun	M - 74	Х	Х				
Green	W - 69	Х	Х				

Table 2. Summary of the main needs

Table 3. Needs per evolving profile level

Level 1	Level 2	Level 3
		Х
Х	Х	Х
		Х
	Х	
Х	Х	
Х	Х	
Х	Х	Х
	Level 1 X X X X X X	Level 1         Level 2           X         X           X         X           X         X           X         X           X         X           X         X           X         X           X         X           X         X           X         X           X         X

for the end of the *OFF* period. To that end, somme relaxation functionalities were included. Table 3 presents the needs that will be applied for each level.

It is important to note that no conflicts were identified during the course of the interviews in the previous activity. This is most likely due to the narrow focus of the interviews, which was specifically on communication needs. Otherwise, it would be advisable to consider applying to other interviews in order to resolve any conflicts that may have arisen, as this may lead to the identification of new requirements.

EICS019:15

During the interviews, the physicians also stated that the system should be adaptable to each user profile level. It was specified that, upon commencing the application, users should respond to a series of questions (based on Well-Being Map questionnaires) regarding their current level of difficulty in communicating. In line with the responses provided, the system will ascertain the user's current level and deliver only the designated functionalities appropriate to that level <sup>4</sup>. Furthermore, it was specified that the system could be adapted by the user, with a number of options available for personalisation.

We concluded that the specific needs of each profile should be specified in detail and the functionalities should be defined to develop a complete software to support communication for PwP. The system may be adapted for each level so that it can be used continuously when the profile evolves to the next level. The main result of this interview was to start the development focusing on profile level 3. With a small technical team, we could not work in parallel for the three levels as desired. Although they say that level 3 corresponds to the minority of PwP, it nonetheless represents those who truly need communicative support. The activities in green in Figure 2 represent this focus. The physicians also decided for the use of tablets as an appropriate technological device. We note that the results of the survey and interviews show that this device is well-known by PwP. Moreover, we decided to use the Android platform to develop a free software application.



Pierre with his wife Thérèse Pierre is a former researcher in the field of marine species biology. He retired four years ago. He worked in a laboratory in northern France. He is 65, married with two children (one boy, one girl).

Pierre has had Parkinson's disease for ten years, but was only diagnosed a year ago by a doctor at Lille University Hospital. Today, he has memory and concentration problems. The volume of his voice is decreasing. Finally, he experiences OFF periods in the morning. In order to reduce the symptoms of the disease, Pierre has to take an increasingly heavy course of medication (with strict timetables to respect).

Pierre's aim is to compensate for the adverse effects of his illness. To do this, he rests during OFF periods. In addition, when he takes part in a discussion, he asks people to wait for his response before starting a new topic. In the presence of strangers, on the other hand, Pierre either doesn't speak or hesitates to speak so as not to reveal his illness.

Pierre communicates regularly with his wife, children and friends, on various topics of daily life. Unfortunately, over the past few months, these exchanges have become increasingly rare, due to difficulties in finding the right words, a slow speech rate and articulation problems.

Pierre had to adapt after the disease was announced and symptoms appeared. However, he has always been in good spirits to continue participating in social life as best he can.

Pierre would like to have a communication aid to enable him to talk more easily with those around him. In addition, such an aid could encourage him to leave his home alone.

#### Characteristics

	Very bad			Very good
Mood	• •	igodol	$oldsymbol{\circ}$	0
Attention/Memory	• •	0	0	0
Voice intensity/Elocution	• •	0	0	0
Upper limb movements	• •	•	0	0

Fig. 7. Simplified example of a defined persona

The specification of the user requirements was performed by applying known techniques. Four personas [55] were defined based on the interviews previously carried out (see a simplified example

<sup>4</sup>This has not yet been implemented, since we have only implemented one level of the evolving profile so far.

in Fig. 7, helping the software designers to keep in mind throughout the software process. We also described the functionalities with an use case diagram and a task analysis model [16].

# 3.4 Design and develop solutions to meet the requirements

This activity consisted of the design of the user interface and code of the software application. It 'started with the development of several mock-ups considering the task models elaborated in the activity *Specify the user requirements*. Co-design is usually recommended in a user-centred design approach to define the user interface layout. Thus, two design workshops were organized with psychologists, psycho-educators, HCI specialists and an expert in communication tools for people with disabilities. These allowed a number of suggestions to be made regarding the functioning and presentation of the system. To that end, we developed different mock-ups for each defined functionality, and carried out other interviews by video-conference in order to validate and define some interactive design options as follows:

- button placement (horizontally aligned at the top or vertically on the side);
- the positioning of the interfaces on the tablet (landscape or portrait);
- the way suggestions are presented to complete the text (text only, pictogram only or pictogram and text);
- the type of keyboard (classic Android keyboard or Swiftkey, commonly used on Android platforms, known for its text prediction quality [4]); and,
- the choice of a symbol for the identification that they are in *OFF* period (thumb down, smile, sad or text).

In addition, we sought to identify whether the chosen features would meet expectations and suggestions for prototype development. Those features concerns to support make a request, communicate with and without vocal synthesiser and relaxation options.

Figure 8 presents three mock-ups about the choice to support text completion with (1) only pictograms, (2) only the text to be completed and, (3) pictograms with text. To that end the PwP interviewed previously were invited to give their opinion and feedback.

A detailed description of the design of the mock-ups and evaluation with PwP can be found in [28]. This user-centred study can be summarized as described below.

*Goal.* To decide on interactive design elements related to the user interface and validate the feature that would be implemented.

*Participants.* Invitation of PwP previously interviewed and contacts from the Parkinson's associations that collaborated with the project.

*Methods.* Presentation by video-conference of the different mock-ups or in person by visiting the PwP at home. This choice was made by the PwP.

**Results.** 16 PwP were interviewed (of which 6 participated in the previous interview, and 3 were previously represented by their relatives, but took part in this phase with a face-to-face interview). The results are as follows:

- the screen presented in landscape (75%);
- the buttons aligned horizontally at the top (63%);
- the use of pictograms followed by word(s) for the proposals to complete the text (50%);
- the thumb for the indication of *OFF* period (44%);
- the swiftkey keyboard (81%);



Fig. 8. Mock-ups for text completion choices

# 3.5 Evaluate Against the Requirements

The activities *Design and develop solutions to meet the requirements* and *Evaluate against the requirements* were performed twice in an iterative way, represented by the green dashed arrow in Figure 2. In the first iteration, a first version of the software was developed based on the previously defined and validated mock-ups. Next, a first evaluation was carried out. Based on the results of this evaluation a second version was developed and then evaluated by the PwP. Therefore, two evaluation studies were carried out. Ethical approval for these studies was obtained from the ethical committees of Lille University (France) and Mons University (Belgium) under the respective references 2021-548-S100 and 10062022EB.

Both studies essentially followed the same research protocol, differing only in the number of participants and the instrument being evaluated (each produced version).

*Goal.* The aim of this study was to evaluate the usability perceived by end users of the communication aid software developed on the basis of the needs identified with PwP and their relatives. The environment (in the context of use) for both studies was in the PwP's home. This was defined to guarantee an evaluation in the context of use that was established as most desirable by the PwP as presented in previous section. In addition, we considered inappropriate to subject PwP to carry out controlled evaluation sessions. Therefore, we planned to leave the PwP to use the application to their convenience over a certain time period (minimum 1 week).

**Participants**. The recruitment method was based on voluntary participation: invitation of the PwP that participated in the previous studies, issuing calls for participation on the Internet and social networks, and contacting Parkinson's associations and hospitals that collaborated with the project. The inclusion criterion was any participant who has been diagnosed with PD, even though the focus of the development was made for a group of PwP at evolving profile level 3 (i.e., those

EICS019:19

with severe communication problems or who no longer speak), from previous interviews, since the vast majority (11/14) requested the need of the developed functionalities (see Table 2).

Methods. The procedure for carrying out the evaluation consisted of four main phases:

- i. *On-site presentation of the software version* A first meeting took place to explain the functionalities of the software, demonstrate it and also provide brief training for its use. Once the information about the software had been clarified, an agreement to take part in the study was required by signing a consent form;
- ii. *Profile data collection* In the same meeting, we collected information on the PwP, regarding age, gender, years since PD diagnosis, and a self-assessment of their difficulty to talk, memory, and level of tremor in the upper limbs (defined based on [71]). To collect this data, we drew up a paper questionnaire which the researcher used to collect the information and filled in as the interview took place;
- iii. Use of the software application The participants were left with a tablet and a detailed user guide of the tool, that was also clearly explained, for a minimum of one week. This period could be extended by user's request. During this time, they were asked to use the software application as an aid to their daily lives. In order to achieve this objective, we prepared a document containing a set of ten scenarios, with a step-by-step explanation of how to perform each one. Furthermore, we established a daily agenda, outlining the minimum number of scenarios to be executed on a daily basis. The ten scenarios are as follows: (1) prepare a text using pictograms; (2) prepare a text and use the vocal synthesizer to communicate with someone; (3) prepare a text, register it and later open it to be used for communication; (4) prepare a text using a predetermined one from the library; (5) modify an existing dialogue and register it; (6) execute an existing text from the library using the synthesizer; (7) perform an urgent request; (8) ask for help; (9) play a piece of music; (10) play a joke. The last two were included as relaxation option.
- iv. *Feedback* A second meeting took place to obtain user feedback on the application, suggestions of improvements, filling out the CSUQ (*Computer System Usability Questionnaire*) questionnaire [45]. To that end, the CSUQ questionnaire was printed, and for each item, users were invited to provide comments regarding their answer. These comments were then written down for further analysis.

Therefore, two visits were made to the PwP's house (one at the beginning and one at the end) in each evaluation. A pre-test was carried out in each evaluation.

**Results**. For the first evaluation, 18 volunteers (PwP) accepted to participate; 2 of them had already participated in a pre-test evaluation; 4 others abandoned the study for medical reasons, but had started using the application (one of them, for instance, sent a report with some suggestions later). Therefore, 12 volunteers (PwP) did the complete evaluation for our analysis (of which 7 had already participated for at least one of the previous studies).

For the second evaluation, 31 volunteers (PwP) accepted to participate; 2 of them had already participated in the pre-test evaluation (of which one had not yet participated in any previous studies), and two others dropped out of the study. Therefore, 27 PwP fully completed the evaluation (of which 12 had already participated for at least one of the previous studies, and 7 of these 12 in the first evaluation).

Table 4 presents the CSUQ final scores considering the evaluation of all participants in both evaluations. Figure 12 shows that the second version obtained a better evaluation for all CSUQ scores, except for the interface quality that remained almost identical. PwP's main criticism of the first version was the use of the native Swiftkey keyboard. The letters were considered too small

and too close together. Including accents was very difficult and, for some PwP, impossible. Some PwP also proposed using an ABCD keyboard instead of the AZERTY (the standard keyboard in the countries where the study was carried out). Another point is that the tablet is very sensitive; for people presenting tremor, it was possible to tap a key twice without meaning to. They suggested introducing, for example, a delay between clicks to avoid typing the same letter twice. Finally, although in the interviews (see Section 3.4) the option for proposition of text completion was to present a pictogram followed by text, they did not like the use of pictograms and asked to use text only.

Figures 9 and 10 show the main screen of the functionality to write a text to support communication of the first and second versions, respectively, where one can note the difference of the keyboard in both versions. A specific keyboard was designed for PwP in the second version, with more separated keys, keys already including accents, and two keyboard types, ABCD and AZERTY. In the second version, the keyboard could be personalised in terms of ABCD/AZERTY options, size of the letters and of the keys. The application screenshot presented in the Figures 9 and 10 have been modified to present the ParkinsonCom application in English. The currently available application languages are French (France and Belgium) and Dutch (Belgium).



Fig. 9. Main screen for preparing a dialog - 1st version

As presented previously, we accepted all volunteers for the evaluation. To better analyse the results considering only the target population for these versions (PwP in level 3 of the evolving profile for communication), we also analysed the data only of PwP with serious communication problems. To achieve this, we clustered participants based on their self assessment on difficulty to talk (greater or equal to 80%) from the Parkinson's Well-Being Map. Table 5 presents this result. We note that in the first evaluation only 2 participants were in the target group. Unfortunately, the 4 participants that abandoned the study for medical reasons were in the target group. Even with this small number, both PwP contributed proposing several suggestions. We note also that, with the exception of the evaluation of interface quality, all other CSUQ scores had a high standard deviation. One of the participants had strong upper limb tremors, which made the use of the keyboard very difficult, and may affect her evaluation. In any case, the keyboard was not considered adequate for the majority of the users in the first version, which justified the design of a specific keyboard for the second version. Figure 13 shows that when analysing the PwP in level 3 (target population for the developed version), the second version obtained a better evaluation score for all items.

A last suggestion from users was to make the "clear text" option more evident and not in the keyboard, since this option (red dustbin button) was very close to the validation (OK) button (see

Human-centred design process when users have an evolving profile

W	rite a o	dialogu	e Th	eme: Dialog	ues withou	t a theme	
Write	e your te	ext here.	•			•••))	
You		lt's	· ·	The	Hello	<b>b</b>	l have
Α	В	С	D	E	F	G	Н
	J	K	L	М	Ν	0	×
Р	Q	R	S	Т	U	V	÷
<b>±</b>	W	X	Y	Ζ	,		â
?123				SPACE			OK

Fig. 10. Main screen for preparing a dialog - 2nd version

v	Vrite a	dialogu	e The	me: Dialo	gues without	a theme	
Write your text here							
						■●))	$\sim$
You		lt's	I T	he	Hello		l have
a	b	С	d	е	f	g	h
i	j	k		m	n	0	р
q	r	S	t	u	V	W	×
<u>+</u>	X	У	Z		?	!	÷
123?	-		SPACE				$\sim$

Fig. 11. Main screen for preparing a dialog - 3rd version

Table 4.	General	CSUQ s	cores
----------	---------	--------	-------

CSUQ Score	1st Eva (n=12)	luation	2nd Evaluation (n=27)		
	Average	Std.	Average	Std.	
Overall (Usability satisfaction)	5.29	1.30	5.60	1.12	
System usefulness	5.36	1.50	5.54	1.49	
Information Quality	5.19	1.36	5.75	1.01	
Interface Quality	5.67	1.01	5.64	0.97	

red and blue buttons in Figure 10. Figure 11 presents the screen for writing a text for dialogue from the latest version of the application. We note that the "clear text" option was included as a new page icon inside the text frame.

CSUQ Score	1st Evaluation (n=2)		2nd Evaluation (n=7)	
	Average	Std.	Average	Std.
Overall (Usability satisfaction)	3.58	2.11	5.71	0.62
System usefulness	3.31	2.31	5.77	1.07
Information Quality	3.47	2.13	5.88	0.49
Interface Quality	5.00	1.00	5.71	0.90

Table 5. CSUQ scores - User profile level 3







Fig. 13. Specific CSUQ comparison

# 3.6 Threats to validity

A fundamental issue related to conducting experiments is the validity of the results. Therefore, to analyse the results of our studies, we considered the four threats of validity proposed by Wohlin et al. [81]: construct validity, internal validity, conclusion validity, and risk to external validity.

Threats to *construct validity* refer to the extent to which the experimental setting actually reflects the construct under study. We mitigated this threat through different actions. For the survey, although we elaborated the questionnaire using closed questions, we included an open item ("others"), to allow the respondent to include other elements, and also evaluate them. Moreover, the questionnaire was reviewed by a physician expert in PD. For the evaluation of prototypes we used a well-known questionnaire with high reliability (CSUQ), and for each answer the respondent was free to explain and make free comments. The first interview was guided by very open questions, which allowed the interviewees to freely express their opinions. Finally, for the second one, we proposed different mock-ups to each item to be investigated.

Threats to *internal validity* draw from influences that can affect the independent variables with respect to causality without the researchers' knowledge. Based on [81], we considered that the factors potentially impacting the studies are related to the selection of participants and their division into different classes and potential special events occurring during the experiment. We did not restrict the selection of participants (PwP) in any one of the study; meaning that we accepted all volunteers that had a diagnosis in PD. However, in the first interview, 4 out of 14 interviewees were the relatives (husband or wife) who answered for their spouse (PwP) due to the latter having strong communication difficulties. Even if the questions were asked with the wording referring to their spouse, a bias may have been present. However, since our goal was to identify the different needs and difficulties regarding communication, we considered that this bias would not affect the result. Therefore, we accepted this threat.

Threats to conclusion validity and external validity are interrelated. The first concerns issues that affect the ability to draw the correct conclusion about the relationships between the treatment and the outcome of an experiment. Threats to external validity, on the other hand, are conditions that limit our ability to generalise the results of the study to other populations. There is always heterogeneity in a study group. If the group is too heterogeneous, there is a risk that the variation due to individual differences will be greater than that due to the treatment. Choosing more homogeneous groups, on the other hand, will affect external validity. For all studies, we did not restrict the choice of participants (PwP). In the first three studies (survey and interviews), this gave us a broader view of a PwP's communication difficulties and needs. We therefore feel that we have minimised the risk of external validity. However, we clearly identified a divergence of needs, perhaps affecting conclusion validity. As our goal was to identify this divergence, we accept this threat. For the last study (software application evaluation), we believe that heterogeneity may affect the conclusion validity of the version developed only for the evolving profile level 3. However, we understand that all participants are aware of the evolution of PD and therefore the degenerative evolution of their profile. Therefore, we preferred to assume this threat. In any case, we also analysed the results considering only the target population answers, even though the number is small.

# 4 DISCUSSION

In this section we present a discussion of the implementation of the ISO process for the context of Parkinson's evolving profiling, seeking to identify lessons learned (Section 4.1). We write these lessons as recommendations in order to define a first version of a human-centred design process presented in Section 4.2.

# 4.1 Lessons Learned

Section 3 described a case study using the adaption of ISO 9241-210 standard process to take into account the evolving profile of PwP in the degradation of their communication capacity. The entire process for developing the communication software aid for PwP took almost three years (from April 2020 to December 2022). During this period, 97 PwP, caregivers (the majority, 4 of 5, PwP's relatives) and health professionals participated in the process taking decisions and giving feedback on the propositions. Table 6 presents the number of participants in each activity. It may appear somewhat unconventional that the user is not involved in the specification phase. This is because the needs were defined based on the transcript of the interviews carried out in the previous activity and no conflicts were detected (as explained in Section 3.2.2). In this activity, the other stakeholders were responsible only for determining the most pertinent needs for each level of the evolving profile.

As we adapted each activity of the standard process to address the evolving profile issues, we noted several important issues. Firstly, defining the different levels of the evolving profile is not straightforward. At the start of this project, we expected that we would be able to identify the different levels by combining the results of a set of typical symptoms. However, with the results of the survey and the interviews, we realised that the evolution of the profile is not similar for all PwP, taking into account their symptoms. This was also confirmed by the physicians working in the project. For this reason, with the support of the physicians, we decided to focus on the self-assessment of PwP in relation to their communication difficulties considering the answer of related items of the Well-being Map questionnaire (that was defined for PwP's own use, see Section 2.2). Second, to better perform human studies with the different stakeholders, we should apply various data collection techniques, such as focus groups and co-design with the end users. However, most of the time of this project was during the COVID-19 pandemic, when these activities could not be carried out. We recall that most PwP are elderly, and considered at risk. We were only able to contact them face to face more than a year after the beginning of the process, taking the necessary precautions and having the whole team vaccinated. Third, with a small technical team and due to the complexity of the potential solutions, we could not address all three levels in parallel as proposed. Then, once we had decided to focus on the third level, we hoped to have more volunteers in the test phase at the target level. In addition, the people identified in this level dropped out of the study due to health problems (unrelated to the project). We also realized that with a degenerative disease profile, the volunteers were evolving during the project period. Users who were classified as level 3 at the start of the project often progressed to other symptoms or diseases, and abandoned the study. Some of them passed away. Others who were classified as level 2 ended the project with great communication difficulties or even being unable to speak. We also noticed that some of those who communicated normally at the beginning lost their ability to communicate by the end.

Despite this, this case study provided a real and complete experience of using the proposed process, bringing with it a number of lessons learned. Some of them were also observed by other studies (e.g. [6, 20, 40, 44, 72, 73]). Based on that, we define a set of recommendations for future projects dealing with the evolving profile of users, as presented below:

I. Start large - try to understand the problem you wish to address at the beginning with a large perspective to better understand the focus of your project. For the initial study (Section 3.2, instead of collecting the opinions of PwP from the project's association partners, an internet survey was carried out, which collected responses from 56 disabled PwP from various geographical locations. At the beginning of the project, we envisioned another type of software system, maybe using cutting-edge technologies. The results of the survey showed

Activity	Study	#PwP	#Health pro- fessionals	#Caregivers
Understand and specify context of use	Initial interview		2	
	Survey	56		
	Structured inter- view	10		4
Specify the user require- ments	Interview		3 <b>(2 new)</b>	1
Design and develop solu- tions to meet the require- ments	Interview	16 <b>(10 new)</b>		
Evaluate against the re- quirements	1st version evalu- ation	18 (2 pre-test, <b>5 new</b> , 4 aban- donments)		
	2nd version eval- uation	31 (2 pre-test, <b>16 new</b> , 2 abandon- ments)	1	
Total		97	5	5

Table 6. Human-centred design in numbers

that we should keep it as simple as possible. The use of a new technology is not trivial when living with a degenerative disease.

- II. Include the target population in the process as much as possible for sure there is no better expert of a specific degenerative disease than those affected by it. We quickly realized that creating a communication aid software without discussing the everyday experience of a PwP would not work. Collecting future-user feedback on the systems's features was seen as precious advice. The co-construction should also be continuous throughout the process to ensure that the adaptations made to the software are in line with what was requested by the target population. As presented in Table 6, 97 PwP participated of the whole process, all of them declare have some difficulty of communication.
- III. Be prepared for users' evolving difficulties during the development process In a project lasting several years, the intensity of symptoms linked to a degenerative disease can increase significantly. This can lead to impaired communication and/or concentration of the people with evolving profile. We need to be prepared for the fact that participants who started out at the beginning of the project may find it increasingly difficult, despite their motivation, to contribute as the project progresses, due to their illness. Some may even find it necessary to withdraw from the project (e.g., 6 abandoned the final evaluation study, Table 6), for reasons beyond their control.
- IV. Accept all volunteers even if it was important to target specific future users to address our study goals, suggestions and feedback can come from all participants, some of whom would have been missed if we have only focused on our target. We note that of the 39 PwPs (12 in the first evaluation and 27 in the second) only 9 (23%, 2 in the first evaluation and 7 in

the second) concern evolving profile level 3. However, several suggestions for improving the application in the evaluation phases came from the other 30 PwP who clearly stated that they evaluated the system taking into account what they know about how they will be in the near future and want to prepare for their future, knowing that their condition will deteriorate.

- V. Use on-site evaluation as much as possible doing the evaluation on-site, in the environment where future users will use the application, and allowing users to use the software when they want to and feel able to, meant that a natural interest in the project and in the use of the software application developed. All 49 evaluations were carried out at the client's home, at a time that was convenient for them. We are convinced that had we opted for a fixed and controlled approach, the results would have been insufficient and we may not have had that number of volunteers.
- VI. Repeat the evaluation process the development of the ParkinsonCom software application was based upon input from repeated evaluations from the future users as recommended by ISO 9241-210. A single evaluation would not have sufficed to achieve the finished product. In some cases, the volunteers tested the product three times and were continuously suggesting improvements. We have performed two iterations only for level 3. To test the whole application addressing all levels, several iterations would be necessary.
- VII. Know when to stop given the nature of an evolving profile and the size of certain disease spectra, such as PD, it may be tempting to fall into the trap of wanting to please everybody. However, constantly adding new features renders software applications more complex for the users and demands higher computational time. This may run the risk of decreased usage and appreciation due to increased complexity. In this case, clearly identify the main needs for each level of the evolving profile. In our experience, as users engaged with the application, they expressed a need for additional functionality that wasn't necessarily focused on communication issues (for example, an agenda to remind them when to take their medication). We began to respond to these needs in order to encourage user interest in the final application. At a certain point it was necessary to stop responding to new requests and focus on the next levels of the evolving user profile.

# 4.2 Towards a Human-centred Design of Interactive Systems for Users with Evolving Profile

Considering the preceding lessons learned and the observation that the evolving profile is a hallmark of any degenerative disease, we formulate a preliminary version of a human-centred design process when users have an evolving profile presented in Figure 14. In this process, the various recommendations presented in the preceding section were integrated. The implementation of this process can be further developed in other projects and improved if deemed necessary. Each activity was reviewed considering the perspective of developing an interactive system for users with an evolving profile, as described below.

**Plan the human-centred process.** In this activity we should describe the activities to be performed, the material and human resources, cost and time constraints. In the case of developing an interactive system for users with an evolving profile, each of these elements must be carefully analysed. As far as human resources are concerned, in addition to technical HCI designers and programmers who are experts in the technology to be used, it is necessary to identify the necessary stakeholders (e.g. physicians, health professionals, caregivers, family members and friends), who are important in helping to understand the users' evolving profile in relation to the problem we want to solve. We must seek out these stakeholders beforehand and make sure they can participate in the whole process. According to [11], in order to distinguish the different types of participants, it



Fig. 14. ISO 9241.210 standard [38] adapted for the design of interactive system centred on users with an evolving profile - the overlapping of activities means their execution for each level of the evolving profile

is important to specify decision-making situations and the groups of participants, communicating who can participate in decisions, with what rights, at what time and on what subject. In addition, we must analyse their constraints in terms of time and local availability to participate. The technology that will be used to solve the problem must also be conscientiously researched in order to best meet the users' needs (be aware that the most modern technologies are not always the most suitable).

Understand and specify context of use. Given the need to develop an interactive system for an evolving profile, this activity should start with a literature review to properly define the problem and understand the disease we will be working with. With a basic knowledge of the subject, one or more interviews with physicians and/or health professionals identified in the previous activity should be carried out. The aim is to confirm understandings obtained from the literature, discuss the difficulties faced by people with an evolving profile and, if possible, potentially ask about the different levels of the evolving profile that could be established. A first identification of potential technology to be used or discarded can also be envisaged. Only after that, we can start to investigate the user needs. In participatory research [11], the mobilised community member has a significant degree of involvement in the project definition process. Based on that, we propose to use a wideranging study method such as surveys in order to identify the user profile with their variability taking into account the evolution of the disease and the focus of the system in question. The idea is to have a general view of the disease in terms of symptoms' impact in daily life, difficulties and needs regarding the problem to be addressed. Interviews can also be applied in a complementary way. These can be carried out with the potential future users, but also their entourage (including caregivers, relatives, friends and medical professionals previously identified) so to gain a deeper and clearer insight into the usability needs linked to the disease. The aim is to identify the different levels of disease progression concerning the problem to be addressed that can be characterised by the user's personal characteristics. These different levels may impact the definition of different contexts of system use and may also require different technological solutions.

*Specify the user requirements.* This activity should be started with one or several structured interviews with the physicians and/or health professionals, in order to present the results obtained in the previous activity, the propositions of evolving profile levels and possible solutions for the envisaged problem. In the event of conflict being detected, it is recommended that new interviews be carried out with user and techniques for managing conflict be applied [65]. The goal in this activity is to validate with the physicians and/or health professionals the propositions in terms of feasibility for the users with the specific evolving profile. In this/these interview(s), discussions about the different functionalities to support the users in each level of the evolving profile can enrich the proposed solutions with new insights from the experience of the physicians and health professionals with the target problem. For each identified level of the evolving profile, the users may have different needs, leading to different functionalities/requirements, which can be considered as different applications within the same software. Figure 14 shows this fact by the drawing of this activity repeated in an overlapping manner. Other interviews can be carried out if necessary to define the functionalities for each level. Considering all the collected data from the survey and interviews, several personas [55] can be written to better specify the needs of the users focusing in the different needs of each level of the evolving profile.

**Design and develop solutions to meet the requirements**. For each proposed specification of a evolving profile level, a design and development of the requirements should be produced. The overlapping of several rectangles in Figure 14 represents the design for each specification produced in the previous activity. This activity requires end users to think through and conceptualize all their ideas as individuals and collectively; they can be accompanied by their family or professional caregivers. The goal is retaining what best corresponds to their needs in terms of user interface design. In this way, participatory design or co-design approaches [67] can be used to actively integrate all the stakeholders, and mainly, the potential end users. In this activity, it is strongly recommended to define and/or validate mock-ups with the users. To achieve this, focus groups or individual interviews with users and stakeholders (caregivers, friends, family, health professionals) may be applied. Depending on whether or not the software development is carried out in parallel to meet the specific needs for each level in the evolving profile, integration and testing of the different components that address each level must be carried out continuously while developing the specific functionalities to address the requirements.

**Evaluate against the requirements**. Once a version of the system is produced, human-computer studies should be carried out with users. Application of usability tests are recommended. Considering the evolving profile, there is a risk that users who participated in the previous activities do not have the same needs since their profile evolved in a degenerative way. We should therefore include new users for evaluating the systems who did not participate in the requirements and design phase. Being aware of that, the evaluation studies in this phase should consider the presentation of the project and the interactive systems being developed. Another aspect to consider is that it can be not easy to submit users to controlled usability-test sessions. In this way, evaluation protocols must take into account the nature of evolving profiles and plan feasible evaluations at a convenient time and in a manner agreed with users. This research protocol should be formally described and approved by ethical committees before performing the evaluation. Considering the different levels of the evolving profile, one or several evaluations should be done for each level (also represented

by the overlapping of rectangles in this activity in Figure 14). Several evaluations are carried out when the results show the need to iterate by doing/improving the design and implementation or by refining the specification (see dashed arrows in Figure 14). It is recommended to use at least one usability questionnaire (see [7] for a review on different standard usability questionnaires) in other to collect the feedback and measure the usability satisfaction. Similar to the previous activity, if the evaluation was done in parallel to meet the specific needs for each level in the evolving profile, the integration and re-valuation of the different versions of the system should be performed.

# 5 CONCLUSION

This article has presented a case study of the definition and application of a human-centred process for the development of interactive systems for people with Parkinson's disease, i.e. users have an evolving profile. This process is supported by a set of user studies carried out from a broadest to most specifics, focused on an identified user profile level. It has taken a step forward by presenting a first version of a generalised software process for the development of software systems when users have an evolving profile.

This article contributes to academia by taking a particular look at the generic process of the ISO 9241-210 standard for the situation when dealing with users with an evolving profile (e.g Parkinson, Alzheimer). In addition, it contributes to practice as it shows step by step how this process was applied in a real case of a large project. Ninety seven (97) people with Parkinson's disease, 5 health professionals (including a physician expert in PD, neurologists and speech therapists) and 5 caregivers (mostly family members) took part in the human-centred design, implementation and validation of ParkinsonCom software application for level 3. In addition, 111 site visits were carried out by the project team (after December 2021) to ensure that the software developed met the needs identified by the target audience as closely as possible. In addition, 165 PwPs have been trained and the ParkinsonCom software has been presented to a thousand people at various scientific, medical and general public events. ParkinsonCom is available on the project's website (https://parkinsoncom.eu) for free download (currently totalling 1,852 downloads<sup>5</sup>).

In future work, we plan to extend the ParkinsonCom software application, to address level 2 of the evolving profile, by proposing functionalities that could capture PwP's voice while speaking and proposing words in real time to complete the sentences when they are blocked due to forgotten words. This involves the use of advanced artificial intelligence algorithms and precise devices for voice detection.

Furthermore, this experience highlighted the importance of defining profile levels at the outset of the process. It is clear that a significant amount of future work is required to define a generic method for specifying evolving profile levels and all the associated attributes that would be useful for characterising each level.

# Acknowledgments

The ParkinsonCom project was developed with the support of the Fonds Européen de Développement Régional Interreg V France-Wallonie-Vlaanderen and the Agence pour une Vie de Qualité (AVIQ), Belgium, to whom the authors would like to express their gratitude. The authors would like to thank all the participants involved in the various phases of the project and the three Parkinson's association - FranceParkinson Nord (France), Vlamseen Parkinson Liga and Association Parinson APk, asbl (Belgium) - without whom this work would not be possible.

<sup>&</sup>lt;sup>5</sup>last access on April 24, 2025

# References

- [1] Wajih Abdallah, Frédéric Vella, Nadine Vigouroux, Adrien Van den Bossche, and Thierry Val. 2019. A Collaborative Talking Assistive Technology for People with Autism Spectrum Disorders. In Universal Access in Human-Computer Interaction. Multimodality and Assistive Environments, Margherita Antona and Constantine Stephanidis (Eds.). Springer International Publishing, Cham, 3–12.
- [2] Silvia Abrahão, Emilio Insfran, Arthur Sluÿters, and Jean Vanderdonckt. 2021. Model-based Intelligent User Interface Adaptation: Challenges and Future Directions. Software and Systems Modeling 20 (10 2021). https://doi.org/10.1007/ s10270-021-00909-7
- [3] Tomás Alves, Joana Natálio, Joana Henriques-Calado, and Sandra Gama. 2020. Incorporating personality in user interface design: A review. *Personality and Individual Differences* 155 (2020), 109709. https://doi.org/10.1016/j.paid. 2019.109709
- [4] Hossam Amer, Joe Osborne, Michael Zaki, and Mohamed Afify. 2024. On-Device Emoji Classifier Trained with GPT-based Data Augmentation for a Mobile Keyboard. *CoRR* abs/2411.05031 (2024).
- [5] Boyd Anderson, Shenggao Zhu, Ke Yang, Jian Wang, Hugh Anderson, Chao Xu Tay, Vincent Y. F. Tan, and Ye Wang. 2018. MANA: Designing and Validating a User-Centered Mobility Analysis System. In Proceedings of the 20th International ACM SIGACCESS Conference on Computers and Accessibility (Galway, Ireland) (ASSETS '18). Association for Computing Machinery, New York, NY, USA, 321–332. https://doi.org/10.1145/3234695.3236340
- [6] Linda Armstrong, Deborah Jans, and Alison MacDonald. 2000. Parkinson's disease and aided AAC: some evidence from practice. *International journal of language & communication disorders* 35, 3 (2000), 377–389. https://doi.org/10. 1080/136828200410636
- [7] Ahlem Assila, Káthia Marçal de Oliveira, and Houcine Ezzedine. 2016. Standardized Usability Questionnaires: Features and Quality Focus. Computer Science and Information Technology 6 (2016). https://api.semanticscholar.org/CorpusID: 54726201
- [8] E. Batselé, H. Geurts, E. Buchet, M-C. Haelewyck, and M. Boutiflat. 2004. Maladie de Parkinson et besoins en matière d'outils technologiques de soutien à la communication: étude qualitative. L'Année Psychologique/Topics in Cognitive Psychology 123 (2004), 81–102. (in press).
- [9] Victoria Batz, H Riess, Mareike Gabele, Dominik Schumacher, and Michael A. Herzog. 2019. CUCKOO Facilitating Communication for People with Mental and Physical Disabilities in Residential Communities. Proceedings of the International Conferences Interfaces and Human Computer Interaction 2019 Game and Entertainment Technologies 2019 and Computer Graphics, Visualization, Computer Vision and Image Processing 2019 (2019). https://api.semanticscholar. org/CorpusID:203049228
- [10] Rongqi Bei, Yajie Liu, Yihe Wang, Yuxuan Huang, Ming Li, Yuhang Zhao, and Xin Tong. 2024. StarRescue: the Design and Evaluation of A Turn-Taking Collaborative Game for Facilitating Autistic Children's Social Skills. In Proceedings of the CHI Conference on Human Factors in Computing Systems (, Honolulu, HI, USA,) (CHI '24). Association for Computing Machinery, New York, NY, USA, Article 67, 19 pages. https://doi.org/10.1145/3613904.3642829
- [11] Jarg Bergold and Stefan Thomas. 2012. Participatory Research Methods: A Methodological Approach in Motion. Forum Qualitative Sozialforschung / Forum: Qualitative Social Research 13, 1 (Jan. 2012). https://doi.org/10.17169/fqs-13.1.1801
- [12] James Berrett, Alison de Kruiff, Sonja Pedell, and Ann Reilly. 2022. Augmented assistive technology: the importance of tailoring technology solutions for people living with dementia at home. *International Journal of Human-Computer Studies* 165 (2022), 102852. https://doi.org/10.1016/j.ijhcs.2022.102852
- [13] Gautam Biswas, Ramkumar Rajendran, Naveeduddin Mohammed, Benjamin S. Goldberg, Robert A. Sottilare, Keith W. Brawner, and Michael Hoffman. 2020. Multilevel Learner Modeling in Training Environments for Complex Decision Making. *IEEE Transactions on Learning Technologies* 13 (2020), 172–185. https://api.semanticscholar.org/CorpusID: 196172369
- [14] Grégory Bourguin, Alain Derycke, and Jean-Claude Tarby. 2001. Beyond the Interface: Co-evolution Inside Interactive Systems – A Proposal Founded on Activity Theory. In *People and Computers XV—Interaction without Frontiers*, Ann Blandford, Jean Vanderdonckt, and Phil Gray (Eds.). Springer London, London, 297–310.
- [15] Diogo Branco, Raquel Bouça, Joaquim Ferreira, and Tiago Guerreiro. 2019. Designing Free-Living Reports for Parkinson's Disease (CHI EA '19). Association for Computing Machinery, New York, NY, USA, 1–6. https://doi. org/10.1145/3290607.3313032
- [16] Sybille Caffiau, Dominique Scapin, Patrick Girard, Mickaël Baron, and Francis Jambon. 2010. Increasing the expressive power of task analysis: Systematic comparison and empirical assessment of tool-supported task models. *Interacting with Computers* 22, 6 (2010), 569–593. https://doi.org/10.1016/j.intcom.2010.06.003 Special Issue on Inclusion and Interaction: Designing Interaction for Inclusive Populations.
- [17] Gaëlle Calvary, Alexandre Demeure, Joëlle Coutaz, and Olfa Daassi. 2004. Adaptation des interfaces homme-machine à leur contexte d'usage Plasticité des IHM. La présentation d'information sur mesure, Numero Special de RIA; Paris, C. et Colineau, N. (editeurs invites). Vol 18 (4) 2004. Date de parution: Septembre 2004. (2004), 577–606.

Proc. ACM Hum.-Comput. Interact., Vol. 9, No. 4, Article EICS019. Publication date: June 2025.

- [18] Murilo C. Camargo, Tathia C. P. Carvalho, Rodolfo M. Barros, Vanessa T. O. Barros, and Matheus Santana. 2019. Improving Usability of a Mobile Application for Children with Autism Spectrum Disorder Using Heuristic Evaluation. In Universal Access in Human-Computer Interaction. Multimodality and Assistive Environments: 13th International Conference, UAHCI 2019, Held as Part of the 21st HCI International Conference, HCII 2019, Orlando, FL, USA, July 26–31, 2019, Proceedings, Part II (Orlando, FL, USA). Springer-Verlag, Berlin, Heidelberg, 49–63. https://doi.org/10.1007/978-3-030-23563-5\_5
- [19] Mauro José Conte, Marcelo Gitirana Gomes Ferreira, and Alejandro R. García Ramírez. 2020. An AAC Mobile-Based Application for People with Intellectual Disability: A Case Study in Brazil. Advances in Human-Computer Interaction 2020 (2020). https://doi.org/10.1155/2020/8932707
- [20] Ana Correia de Barros, João Cevada, Àngels Bayés, Sheila Alcaine, and Berta Mestre. 2013. User-centred design of a mobile self-management solution for Parkinson's disease. In *Proceedings of the 12th International Conference on Mobile and Ubiquitous Multimedia*. Association for Computing Machinery, New York, NY, USA, Article 23, 10 pages.
- [21] Káthia Marçal de Oliveira, Elise Batselé, Sophie Lepreux, Elise Buchet, Christophe Kolski, Mathilde Boutiflat, Véronique Delcroix, Hélène Geurts, Kodzo Apedo, Loïc Dehon, Houcine Ezzedine, Yohan Guerrier, Marie-Claire Haelewyck, Nicolas Jura, Philippe Pudlo, and Yosra Rekik. 2021. ParkinsonCom Project: Towards a Software Communication Tool for People with Parkinson's Disease. In Universal Access in Human-Computer Interaction. Design Methods and User Experience, Margherita Antona and Constantine Stephanidis (Eds.). Springer International Publishing, Cham, 418–428.
- [22] Katherine H O Deane, Helen Flaherty, David J Daley, Roland Pascoe, Bridget Penhale, Carl E Clarke, Catherine Sackley, and Stacey Storey. 2014. Priority setting partnership to identify the top 10 research priorities for the management of Parkinson's disease. *BMJ Open* 4, 12 (2014). https://doi.org/10.1136/bmjopen-2014-006434 arXiv:https://bmjopen.bmj.com/content/4/12/e006434.full.pdf
- [23] Gilles Fénelon, Florence Mahieux, Renaud Huon, and Marc Ziégler. 2000. Hallucinations in Parkinson's disease: prevalence, phenomenology and risk factors. Brain 123, 4 (2000), 733–745. https://doi.org/10.1093/brain/123.4.733
- [24] Guangshang Gao. 2019. A Survey of User Profiles Methods. Data Analysis and Knowledge Discovery 3, 3 (2019), 25 35. https://doi.org/10.11925/infotech.2096-3467.2018.0784
- [25] Jing Gao, Feng Tian, Junjun Fan, Dakuo Wang, Xiangmin Fan, Yicheng Zhu, Shuai Ma, Jin Huang, and Hongan Wang. 2018. Implicit Detection of Motor Impairment in Parkinson's Disease from Everyday Smartphone Interactions. In *Extended Abstracts of the 2018 CHI Conference on Human Factors in Computing Systems* (Montreal QC, Canada) (*CHI EA* '18). Association for Computing Machinery, New York, NY, USA, 1–6. https://doi.org/10.1145/3170427.3188502
- [26] Alberto Gaspar, Miriam Gil, José Ignacio Panach, and Verónica Romero. 2024. Towards a general user model to develop intelligent user interfaces. *Multimedia Tools and Applications* (2024), 1573–7721. https://doi.org/10.1007/s11042-024-18240-w
- [27] R. C. Gibson, M. M. Bouamrane, and M. Dunlo. 2019. Design Requirements for a Digital Aid to Support Adults With Mild Learning Disabilities During Clinical Consultations: Qualitative Study With Experts. *JMIR rehabilitation and* assistive technologies 6, 1 (2019), e10449. https://doi.org/10.2196/10449
- [28] Yohan Guerrier, Sophie Lepreux, Christophe Kolski, Véronique Delcroix, and Káthia Marçal de Oliveira. 2022. Human-Computer Interface Design of a Communication Aid Software System for People with Parkinson's Disease. In Proceedings of the Workshops on Engineering Interactive Computing Systems (EICS-WS 2022) co-located with the 14th ACM SIGCHI Symposium on Engineering Interactive Computing Systems (SIGCHI 2022), Sophia Antipolis, France, June 21, 2022 (CEUR Workshop Proceedings, Vol. 3404), Thomas Kosch (Ed.). CEUR-WS.org, 68–74. https://ceur-ws.org/Vol-3404/paper8.pdf
- [29] Yohan Guerrier, Káthia Marçal de Oliveira, Chirstophe Kolski, Sophie Lepreux, Kodzo Apedo, Veronique Delcroix, and Houcine Ezzedine. 2021. Une étude systématique pour la conception d'un système d'aide à la communication pour les personnes atteintes de la Maladie de Parkinson. In Actes du 39ème Congrès INFORSID, INFormatique des Organisations et Systèmes d'Information et de Décision (Dijon, France). 71–86.
- [30] Marine Guffroy, Nadine Vigouroux, Christophe Kolski, Frédéric Vella, and Philippe Teutsch. 2017. From Human-Centered Design to Disabled User & Ecosystem Centered Design in Case of Assistive Interactive Systems. Int. J. Sociotechnology Knowl. Dev. 9, 4 (2017), 28–42. https://doi.org/10.4018/IJSKD.2017100103
- [31] Raquel Hervás, Virginia Francisco, Gonzalo Méndez, and Susana Bautista. 2019. A User-Centred Methodology for the Development of Computer-Based Assistive Technologies for Individuals with Autism. In *Human-Computer Interaction* – *INTERACT 2019*, David Lamas, Fernando Loizides, Lennart Nacke, Helen Petrie, Marco Winckler, and Panayiotis Zaphiris (Eds.). Springer International Publishing, Cham, 85–106.
- [32] Margaret M. Hoehn and Melvin D. Yahr. 1967. Parkinsonism: onset, progression and mortality. Neurology 17, 5 (1967), 427–427. https://doi.org/10.1212/WNL.17.5.427
- [33] Naotsune Hosono, Fumihiro Miyajima, Toshiyuki Inaba, Masaru Nishijima, Michio Suzuki, Hiroyuki Miki, and Yutaka Tomita. 2013. The Urgent Communication System for Deaf and Language Dysfunction People. In Human Interface and the Management of Information. Information and Interaction for Health, Safety, Mobility and Complex Environments, Sakae Yamamoto (Ed.). Springer Berlin Heidelberg, Berlin, Heidelberg, 269–274.

- [34] Tomoka Ikeda, Masakazu Hirokawa, and Kenji Suzuki. 2020. A Multimodal Communication Aid for Persons with Cerebral Palsy Using Head Movement and Speech Recognition. Springer-Verlag, Berlin, Heidelberg, 429–436. https: //doi.org/10.1007/978-3-030-58805-2-51
- [35] Silvia Imbesi and Giuseppe Mincolelli. 2020. Monitoring Older People: An Overview of Devices Responding to Significant Needs of Elderly Affected by Parkinson's Disease. In Advances in Additive Manufacturing, Modeling Systems and 3D Prototyping, Massimo Di Nicolantonio, Emilio Rossi, and Thomas Alexander (Eds.). Springer International Publishing, Cham, 646–656.
- [36] Lizeth Islas, Víctor M. González, and Marcelo Mejía. 2013. Developing a Mobile Application for Language Disabled Children with User Centered Design. In *HCI International 2013 - Posters' Extended Abstracts*, Constantine Stephanidis (Ed.). Springer Berlin Heidelberg, Berlin, Heidelberg, 236–240.
- [37] ISO 9241-11:2018 2018. Ergonomics of human-system interaction Part 11: Usability: Definitions and concepts. Standard. International Organization for Standardization, Geneva, CH.
- [38] ISO 9241-210:2019 2019. Ergonomics of human-system interaction Part 210: Human-centred design for interactive systems. Standard. International Organization for Standardization, Geneva, CH.
- [39] Anthony Bokolo Jnr. 2024. Examining the Use of Intelligent Conversational Voice-Assistants for Improved Mobility Behavior of Older Adults in Smart Cities. *International Journal of Human–Computer Interaction* 0, 0 (2024), 1–22. https://doi.org/10.1080/10447318.2024.2344145 arXiv:https://doi.org/10.1080/10447318.2024.2344145
- [40] Jade Kandel, Chelsea Duppen, Qian Zhang, Howard Jiang, Angelos Angelopoulos, Ashley Paula-Ann Neall, Pranav Wagh, Daniel Szafir, Henry Fuchs, Michael Lewek, and Danielle Albers Szafir. 2024. PD-Insighter: A Visual Analytics System to Monitor Daily Actions for Parkinson's Disease Treatment. In *Proceedings of the CHI Conference on Human Factors in Computing Systems* (, Honolulu, HI, USA,) (*CHI '24*). Association for Computing Machinery, New York, NY, USA, Article 130, 18 pages. https://doi.org/10.1145/3613904.3642215
- [41] Azriel Kaplan, Shirel Barkan-Slater, Yair Zlotnik, and Shelly Levy-Tzedek. 2024. Robotic technology for Parkinson's disease: Needs, attitudes and concerns of individuals with Parkinson's disease and their family members. A focus group study. International Journal of Human-Computer Studies 181 (2024), 103148. https://doi.org/10.1016/j.ijhcs.2023.103148
- [42] Jan Patrick Kopetz, Svenja Burgsmüller, Ann-Kathrin Vandereike, Michael Sengpiel, Daniel Wessel, and Nicole Jochems. 2019. Finding User Preferences Designing the Innovative Interaction Device "BIRDY" for Intensive Care Patients. In Proceedings of the 20th Congress of the International Ergonomics Association (IEA 2018), Sebastiano Bagnara, Riccardo Tartaglia, Sara Albolino, Thomas Alexander, and Yushi Fujita (Eds.). Springer International Publishing, Cham, 698–707.
- [43] Elina Kuosmanen, Eetu Huusko, Niels van Berkel, Francisco Nunes, Julio Vega, Jorge Goncalves, Mohamed Khamis, Augusto Esteves, Denzil Ferreira, and Simo Hosio. 2023. Exploring crowdsourced self-care techniques: A study on Parkinson's disease. *International Journal of Human-Computer Studies* 177 (2023), 103062. https://doi.org/10.1016/j. ijhcs.2023.103062
- [44] Elina Kuosmanen, Valerii Kan, Aku Visuri, Simo Hosio, and Denzil Ferreira. 2020. Let's Draw: Detecting and Measuring Parkinson's Disease on Smartphones. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems (Honolulu, HI, USA) (CHI '20). Association for Computing Machinery, New York, NY, USA, 1–9. https: //doi.org/10.1145/3313831.3376864
- [45] James R. Lewis. 2018. Measuring Perceived Usability: The CSUQ, SUS, and UMUX. International Journal of Human–Computer Interaction 34, 12 (2018), 1148–1156. https://doi.org/10.1080/10447318.2017.1418805
- [46] Fang Suey Lin, Ching-Yi Lin, Yu Jie Hsueh, Chun-Yi Lee, and Chun-Pei Hsieh. 2016. Graphical Tools for Doctor-Patient Communication: An App Prototype Design in Children's Pain Management. In HCI International 2016 – Posters' Extended Abstracts, Constantine Stephanidis (Ed.). Springer International Publishing, Cham, 318–323.
- [47] Yuanhua Liu, Anna-Lisa Osvalder, and Marianne Karlsson. 2010. Considering the Importance of User Profiles in Interface Design. https://doi.org/10.5772/8903
- [48] Sinziana Mazilu, Ulf Blanke, Michael Hardegger, Gerhard Tröster, Eran Gazit, and Jeffrey M. Hausdorff. 2014. GaitAssist: a daily-life support and training system for parkinson's disease patients with freezing of gait. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems* (Toronto, Ontario, Canada) (*CHI '14*). Association for Computing Machinery, New York, NY, USA, 2531–2540. https://doi.org/10.1145/2556288.2557278
- [49] Roisin McNaney, Ivan Poliakov, John Vines, Madeline Balaam, Pengfei Zhang, and Patrick Olivier. 2015. LApp: A Speech Loudness Application for People with Parkinson's on Google Glass. In Proceedings of the 33rd Annual ACM CHI Conference on Human Factors in Computing Systems (Seoul, Republic of Korea) (CHI '15). Association for Computing Machinery, New York, NY, USA, 497–500. https://doi.org/10.1145/2702123.2702292
- [50] Roisin McNaney, Emmanuel Tsekleves, and Jonathan Synnott. 2020. Future Opportunities for IoT to Support People with Parkinson's. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems (Honolulu, HI, USA) (CHI '20). Association for Computing Machinery, New York, NY, USA, 1–15. https://doi.org/10.1145/3313831.3376871
- [51] Roisin McNaney, John Vines, Andy Dow, Harry Robinson, Heather Robinson, Kate McDonald, Leslie Brown, Peter Santer, Don Murray, Janice Murray, David Green, and Peter Wright. 2018. Enabling the Participation of People with

Parkinson's and their Caregivers in Co-Inquiry around Collectivist Health Technologies. In *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems* (Montreal QC, Canada) (*CHI '18*). Association for Computing Machinery, New York, NY, USA, 1–14.

- [52] Sonali R. Mishra, Predrag Klasnja, John MacDuffie Woodburn, Eric B. Hekler, Larsson Omberg, Michael Kellen, and Lara Mangravite. 2019. Supporting Coping with Parkinson's Disease Through Self Tracking (CHI '19). Association for Computing Machinery, New York, NY, USA, 1–16. https://doi.org/10.1145/3290605.3300337
- [53] Olfa Nasraoui, Maha Soliman, Esin Saka, Antonio Badia, and Richard Germain. 2008. A Web usage mining framework for mining evolving user profiles in dynamic Web sites. *IEEE Transactions on Knowledge and Data Engineering* 20, 2 (2008), 202 – 215. https://doi.org/10.1109/TKDE.2007.190667
- [54] Tembalethu Ndwe, Ndze D. Jumbam, and Naomi Isabirye. 2017. UCD Pre-design Research Methods in ICT for Development. In The 11th International Multi-Conference on Society, Cybernetics and Informatics (IMSCI 2017). 67–72.
- [55] Lene Nielsen. 2019. Personas User Focused Design. Publishing Company, Incorporated.
- [56] Donald A. Norman and Stephen W. Draper. 1986. User-Centered System Design: New Perspectives on Human-Computer Interaction. Lawrence Earlbaum Associates, Hillsdale, NJ.
- [57] Francisco Nunes and Geraldine Fitzpatrick. 2015. Self-Care Technologies and Collaboration. International Journal of Human-Computer Interaction 31, 12 (2015), 869–881. https://doi.org/10.1080/10447318.2015.1067498 arXiv:https://doi.org/10.1080/10447318.2015.1067498
- [58] Movement Disorder Society Task Force on Rating Scales for Parkinson's Disease. 2003. The unified Parkinson's disease rating scale (UPDRS): status and recommendations. *Movement Disorders* 18, 7 (2003), 738–750. https://doi.org/10.1002/mds.10473
- [59] Stuart Palmer. 2001. Engineering flexible teaching and learning in engineering education. European Journal of Engineering Education 26, 1 (2001), 1–13. https://doi.org/10.1080/03043790010009149
- [60] Paul van Schaik Philip Barker and Spencer Hudson. 1998. Mental Models and Lifelong Learning. Innovations in Education and Training International 35, 4 (1998), 310–318. https://doi.org/10.1080/1355800980350406
- [61] E. Reinertsen and G. D.. Clifford. 2017. Les signes non moteurs de la maladie de Parkinson. La Presse Médicale 46, 2 (2017), 195–201.
- [62] E. Reinertsen and G. D., Clifford. 2018. A review of physiological and behavioral monitoring with digital sensors for neuropsychiatric illnesses. *Physiological measurement* 39, 5 (2018), 05TR01. https://doi.org/10.1111/jnc.13691
- [63] Rizdania Rizdania, Herman Tolle, and Ismiarta Aknuranda. 2018. Design and Usability Evaluation of Communication Board for Deaf People with User-Centered Design Approach. *International Journal of Interactive Mobile Technologies* (*iJIM*) 12 (03 2018), 197. https://doi.org/10.3991/ijim.v12i2.8100
- [64] Bill Schilit, Norman Adams, and Roy Want. 1994. Context-Aware Computing Applications. In Proc. of The IEEE Workshop on Mobile Computing Systems and Applications. 85 – 90. https://doi.org/10.1109/WMCSA.1994.16
- [65] Naude Scribante, Leon Pretorius, and Siebert Benade. 2016. Conflict in the Requirements Engineering Process. In Proceedings of the 12th INCOSE SA Systems Engineering.
- [66] Ben Shneiderman. 1987. Designing the User Interface: Strategies for Effective Human-Computer Interaction. Addison-Wesley Publishing Co, Reading, MA.
- [67] Jesper Simonsen and Toni Robertson. 2013. Routledge International Handbook of Participatory Design. Taylor & Francis group.
- [68] S. Sveinbjornsdottir. 2016. The clinical symptoms of Parkinson's disease. Journal of neurochemistry 139, 1 (2016), 318–324. https://doi.org/10.1111/jnc.13691
- [69] Feng Tian, Xiangmin Fan, Junjun Fan, Yicheng Zhu, Jing Gao, Dakuo Wang, Xiaojun Bi, and Hongan Wang. 2019. What Can Gestures Tell? Detecting Motor Impairment in Early Parkinson's from Common Touch Gestural Interactions. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems (Glasgow, Scotland Uk) (CHI '19). Association for Computing Machinery, New York, NY, USA, 1–14. https://doi.org/10.1145/3290605.3300313
- [70] Andrea C. Tricco, Etienne V. Langlois, and Sharon E. Straus. 2017. *Rapid reviews to strengthen health policy and systems: a practical guide.* World Health Organization.
- [71] UCB. [n. d.]. Parkinson's Wellbeing map<sup>™</sup>. https://www.ucb.com/patients/Support-tools/Parkinson-s-Well-Being-Map. Accessed: 2025-05-07.
- [72] Jenny Vafeiadou, Asimina Vasalou, and George Roussos. 2021. Self-tracking in Parkinson's The Lived Efforts of Self-management. Proc. ACM Hum.-Comput. Interact. 5, CSCW1, Article 84 (April 2021), 25 pages. https://doi.org/10. 1145/3449158
- [73] Julio Vega, Samuel Couth, Ellen Poliakoff, Sonja Kotz, Matthew Sullivan, Caroline Jay, Markel Vigo, and Simon Harper. 2018. Back to Analogue: Self-Reporting for Parkinson's Disease. In *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems* (Montreal QC, Canada) (*CHI '18*). Association for Computing Machinery, New York, NY, USA, 1–13. https://doi.org/10.1145/3173574.3173648

- [74] Elizabeth S. Veinott. 2022. Adaptive Instruction: A Case Study of Gamified Practice and Testing in a Large Introductory Psychology Course. In HCI International 2022 - Late Breaking Papers. Interaction in New Media, Learning and Games: 24th International Conference on Human-Computer Interaction, HCII 2022, Virtual Event, June 26–July 1, 2022, Proceedings. Springer-Verlag, Berlin, Heidelberg, 455–467. https://doi.org/10.1007/978-3-031-22131-6\_34
- [75] John Vines, Róisín McNaney, Amey Holden, Ivan Poliakov, Peter Wright, and Patrick Olivier. 2016. Our Year With the Glass: Expectations, Letdowns and Ethical Dilemmas of Technology Trials With Vulnerable People. Interacting with Computers 29, 1 (12 2016), 27–44. https://doi.org/10.1093/iwc/iww017 arXiv:https://academic.oup.com/iwc/articlepdf/29/1/27/8508333/iww017.pdf
- [76] Joaquin A. Vizcarra, Álvaro Sánchez-Ferro, Walter Maetzler, Luca Marsili, Lucia Zavala, Anthony E. Lang, Pablo Martinez-Martin, Tiago A. Mestre, Ralf Reilmann, Jeffrey M. Hausdorff, E. Ray Dorsey, Serene S. Paul, Judith W. Dexheimer, Benjamin D. Wissel, Rebecca L. M. Fuller, Paolo Bonato, Ai Huey Tan, Bastiaan R. Bloem, Catherine Kopil, Margaret Daeschler, Lauren Bataille, Galit Kleiner, Jesse M. Cedarbaum, Jochen Klucken, Aristide Merola, Christopher G. Goetz, Glenn T. Stebbins, Alberto J. Espay, and on behalf of the MDS Technology Task Force and the MDS Rating Scales Program Electronic Development Ad-Hoc Committee . 2019. The Parkinson's disease e-diary: Developing a clinical and research tool for the digital age. *Movement Disorders* 34, 5 (2019), 676–681. https://doi.org/10.1002/mds.27673
- [77] Torben Volkmann, Deniz Akyildiz, Nikolas Knickrehm, Fabian Vorholt, and Nicole Jochems. 2020. Active Participation of Older Adults in the Development of Stimulus Material in an Storytelling Context. In Human Aspects of IT for the Aged Population. Technologies, Design and User Experience: 6th International Conference, ITAP 2020, Held as Part of the 22nd HCI International Conference (Copenhagen, Denmark). Springer-Verlag, Berlin, Heidelberg, 84–95. https: //doi.org/10.1007/978-3-030-50252-2
- [78] Torben Volkmann, Friedemann Dohse, Michael Sengpiel, and Nicole Jochems. 2019. Age-Appropriate Design of an Input Component for the Historytelling Project. In *Proceedings of the 20th Congress of the International Ergonomics Association (IEA 2018)*, Sebastiano Bagnara, Riccardo Tartaglia, Sara Albolino, Thomas Alexander, and Yushi Fujita (Eds.). Springer International Publishing, Cham, 672–680.
- [79] Francesco Vona, Emanuele Torelli, Eleonora Beccaluva, and Franca Garzotto. 2020. Exploring the Potential of Speechbased Virtual Assistants in Mixed Reality Applications for People with Cognitive Disabilities. In Proceedings of the International Conference on Advanced Visual Interfaces (Salerno, Italy) (AVI '20). Association for Computing Machinery, New York, NY, USA, Article 21, 9 pages. https://doi.org/10.1145/3399715.3399845
- [80] E. J. Weisblatt, C. S. Langensiepen, B. Cook, C. Dias, K. Plaisted Grant, M. Dhariwal, M. S. Fairclough, S. E. Friend, A. E. Malone, B. Varga-Elmiyeh, A. Rybicki, P. Karanth, and M. K. Belmonte. 2019. A Tablet Computer-Assisted Motor and Language Skills Training Program to Promote Communication Development in Children with Autism: Development and Pilot Study. *International Journal of Human–Computer Interaction* 35, 8 (2019), 643–665. https: //doi.org/10.1080/10447318.2018.1550176 arXiv:https://doi.org/10.1080/10447318.2018.1550176
- [81] Claes Wohlin, Per Runeson, Martin Höst, Magnus C. Ohlsson, Björn Regnell, and Anders Wesslén. 2012. Experimentation in Software Engineering. Springer-Berlin Heidelberg.
- [82] Alan Michael Woods, Mariusz Nowostawski, Elizabeth A Franz, and Martin Purvis. 2014. Parkinson's disease and essential tremor classification on mobile device. *Pervasive and Mobile Computing* 13 (2014), 1–12. https://doi.org/10. 1016/j.pmcj.2013.10.002
- [83] Lian Zhang, Qiang Fu, Amy Swanson, Amy Weitlauf, Zachary Warren, and Nilanjan Sarkar. 2018. Design and Evaluation of a Collaborative Virtual Environment (CoMove) for Autism Spectrum Disorder Intervention. ACM Trans. Access. Comput. 11, 2, Article 11 (jun 2018), 22 pages. https://doi.org/10.1145/3209687